Author’s response to reviews

Title: Rethinking obstetric violence and the “neglect of neglect”: the silence of a Labour Ward milieu in a South African district hospital

Authors:

Maura Lappeman (mauralappeman@gmail.com)
Leslie Swartz (lswartz@sun.ac.za)

Version: 2 Date: 20 Aug 2019

Author’s response to reviews:

FORMATTING CHANGES:

- Name and email addresses of Authors (title page)- Changed
- COREQ guidelines- Please see Attached
- Names should be stated as initials in Authors contribution- Changed
- Rename Introduction to Background- Corrected
- Rename Methodology to Methods- Corrected